Author's response to reviews

Title: Taking Action Together: A YMCA-Based Protocol to Prevent Type-2 Diabetes in High-BMI Inner-City African American Children

Authors:

Lorrene D Ritchie (lorrene_ritchie@sbcglobal.net)
Sushma Sharma (sushma@berkeley.edu)
Joanne P Ikeda (iikeda@berkeley.edu)
Rita A Mitchell (rita@ritamitchell.org)
Aarthi Raman (aarthi.raman2@gmail.com)
Barbara S Green (bsgreen@berkeley.edu)
Mark L Hudes (mlhudes@hotmail.com)
Sharon E Fleming (sfleming@berkeley.edu)

Version: 7 Date: 7 May 2010

Author's response to reviews:

Formatting changes ONLY were made as requested:

-----------------------------

Major revisions (we require the author to make these changes) - NO MAJOR CHANGES WERE MADE.

Trial registration: if this is a clinical trial please provide the trial registration number. THE REGISTRATION NUMBER HAS BEEN ADDED (SEE ABSTRACT PAGE) AS REQUESTED.

References: please provide futher citation details for reference 46. THIS REFERENCE WAS DELETED.

You can view the BMC reference guide at the following link

http://www.biomedcentral.com/bmccgenet/ifora/#references. SOME CITATIONS WERE EDITED TO ENSURE FORMATTING WAS IN REQUIRED STYLE.

Please make sure requested changes are made to the latest formatted version of your manuscript. DONE!